Author’s response to reviews

Title: Fluoroquinolone consumption and Escherichia coli resistance in Japan: An ecological study

Authors:

Fumitaka Terahara (t.fumitaka.k.k@gmail.com)

Hiroshi Nishiura (nishiurah@gmail.com; nishiurah@med.hokudai.ac.jp)

Version: 2 Date: 02 Apr 2019

Author’s response to reviews:

3 April 2019

Dr Megan Barling

Associate Editor

BMC Public Health

Re: Fluoroquinolone consumption and Escherichia coli resistance in Japan: An ecological study (PUBH-D-18-03077R1)

Herewith, we wish to submit the revised manuscript (PUBH-D-18-03077R1) by Fumitaka Terahara and Hiroshi Nishiura for consideration for publication as an Original Article in BMC Public Health.

We have revised the manuscript in accordance with the editorial comments.

We confirm that the map depicted in Figure 1 is our original. There is no copyright protection.

All authors have read and approved the final version of the manuscript. We have no conflicts of interest. We hereby approve this manuscript for publication, declare that the manuscript has not been considered for publication elsewhere, and that these results have never been reported. We
hope that the changes are satisfactory and that you now consider the revised manuscript suitable for publication.

Yours sincerely

Hiroshi Nishiura

Professor of Hygiene
Graduate School of Medicine
Hokkaido University, Sapporo, Japan